EO-1 Weekly Status Week of Jan 13 – Jan 19, 2006
Day of Year 013-019
Mission Day 1880 - 1886
EO-1 General:

The EO-1 satellite performed nominally this week.  EO-1 averaged approximately 12 Data Collection Events (DCE’s) per day for the week for a total of 83 DCE’s acquired and recovered.
In addition, the instruments underwent calibration and decontamination cycling as follows:

· An all instrument lunar calibration was performed on 01/14/2006 at 22:04z (06-014).

· The HSI was power cycled for PPT testing from 01/18/2006 (06-018) at 19:20z to 01/19/2006 (06-019) at 14:25z.  The cryo-cooler was restarted on 01/19/2006 (06-019) at 19:30z.
· ALI performed an outgassing from 01/18/2006 (06-018) at 09:30z to 01/19/2006 (06-019) at 19:10z.
Spacecraft:

All spacecraft subsystems performed nominally this week.  

Pulsed Plasma Thruster (PPT)

A PPT test was performed on 01/18/2006 (06-018) to check out PPT operations using a higher minimum charge cycles.  Prior to the test ACS Table 93 was updated and uplinked to increase the minimum charge cycles from 3 to 6.  During the test the PPT saturated twice while attempting to fire spark plug #1.  Analysis of the PPT saturations is under investigation.

Electrical Power Subsystem (EPS)

A solar array characterization test was performed on 01/18/2006 at 20:30z (06-018).
Instruments:

The instruments operated nominally this week. 

Technology Activities:
The EO-1 extended mission phase has been transformed into an on-orbit testbed for advanced technology and hyperspectral research.  The status of various validation efforts is as follows:

Sensorweb & Virtual Observatory Demonstrations

The automated schedule generation software continues to operate smoothly producing all weekly and daily activity schedules without problems.  All priority replacements were uplinked automatically without operator intervention.  All sensor web replacements (triggered scenes from other sensor installations) were also uplinked automatically.

Autonomous Sciencecraft Experiment (ASE)

ASE was shut down during the PPT test and this time was used to clear the CASPER schedule.  The clearing of the CASPER schedule was performed to release locked memory and will need to be performed about every 4 months.
Hybrid Ground Phased Array Antenna Validation (funded & in process) 

We received the three newly fabricated Space Fed Lens units.  Sebastienne Rondineau, from the Univ. of Colorado, assembled the units at the Glenn Research Center.  The initial unit was tested and the characteristics were as expected.  Testing will continue for the other units through the next week.  The initial unit will be shipped to Georgia Tech in preparation for our demonstration in the March timeframe. 
Onboard and Ground-based Compression of Hyperspectral Data (unfunded & in definition)

Nothing new to report.

Precision Attitude Control Using the Three Axis Magnetometer (TAM) (unfunded and in definition) 
Nothing new to report.
Livingstone Onboard Model Based Diagnostic Tools & Autonomy Testbed

Nothing new to report.
Ground and Space Network:

EOS Data and Operations System (EDOS), GSFC, MD, USA

EDOS continues to provide X-Band data transfer from both WGS and SGS to the EO-1 Mission Operations Center (MOC).
Wallops Ground Station (WGS), Wallops Island, VA, USA 

There are no major problems to report.
DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.
Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.

Svalbard Ground Station (SGS), Longyearbyen, Norway

There are no major problems to report.
Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There were no major problems to report.

USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA

LGS is now being scheduled only for contingencies.

Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

There are no major problems to report.

Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

· MLT control burn on 01/31/2006 (06-031)

· MLT control burn on 02/14/2006 (06-044)

Imagery Status:

Scenes and Engineering Cals Planned for week of Jan 13, 2006 – Jan 19, 2006                83
# of scenes recovered for week of Jan 13 – Jan 19, 2006                                                    83
Total scenes and engineering calibrations planned for entire mission        approx 26,003
Total Scenes:  ALI scenes in the Level 0 archive            24,710 (as of Jan 19, 2006)

                        Hyperion scenes in the Level 0 archive    24,473
Publications and Presentations Status (as of 01-19-06):

337 publications 

233 external presentations

  47 articles and press releases
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